# RESEARCH

# **Reproductive Health**



# The prevalence of unintended pregnancy and its influence on pregnancy experience in Tabriz, Iran, 2023: a cross-sectional study

Mahsa Maghalian<sup>1</sup>, Roghayeh Nikanfar<sup>2</sup>, Mahsan Nabighadim<sup>3</sup> and Mojgan Mirghafourvand<sup>4\*</sup>

# Abstract

**Background** There is a lack of quantitative studies that specifically measure the association between the experience of pregnancy and unintended pregnancy. The present study aims to address the prevalence of unintended pregnancy and identify its predictors. Additionally, the study explores whether unintended pregnancy is associated with pregnancy uplifts and hassles.

**Methods** This cross-sectional study was conducted on 488 pregnant women between 20 to 40 weeks' gestation at the comprehensive health center in Tabriz City from February 2022 to January 2023. A cluster sampling method was used for sampling, and data were collected using socio-demographic questionnaires and the Pregnancy Experience Scale (PES). Descriptive statistics were used to describe the socio-demographic characteristics and the prevalence of unintended pregnancy. Binary logistic regression was employed to identify the predictors of pregnancy desirability. To examine the relationship between unintended pregnancy and pregnancy experience, an independent t-test was used for bivariate analysis, and a general linear model (GLM) was utilized for multivariate analysis, with control for potential confounding variables.

**Results** The prevalence of unintended pregnancies was 30.7% (24.3% unwanted pregnancies, and 6.4% mistimed pregnancies). The results of the binary logistic regression indicated that the lower age of both the woman and her spouse were significant predictors for unintended pregnancy (P < 0.05). Based on an independent t-test, the mean score for uplifts in women with unintended pregnancy was significantly lower than in women with intended pregnancy (mean difference (MD): -4.99; 95% confidence interval (CI): -5.96 to -4.02; p < 0.001), While the mean score of hassles in women with unintended pregnancy was significantly higher than women with intended pregnancy (MD: 2.92; 95% CI: 2.03 to 3.80; p < 0.001). The results of GLM showed that women who had unintended pregnancies had significantly lower scores for uplifts (B=-4.99; 95% CI: -5.96 to -4.03; P < 0.001) and higher scores for hassles (B=2.92; 95% CI: 2.06 to 3.78; P < 0.001).

**Conclusions** The high prevalence of unintended pregnancies in Tabriz highlights the importance of targeted interventions to address this issue, considering the policy framework and unique challenges faced by women. Future studies should focus on developing context-specific interventions that effectively meet the needs of women with unintended pregnancies.

Keywords Unplanned pregnancy, Prevalence, Experience, Predictors, Association, Cross-sectional study

\*Correspondence: Mojgan Mirghafourvand mirghafourvand@gmail.com Full list of author information is available at the end of the article



© The Author(s) 2024. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.gr/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.gr/licenses/by/4.0/.

# **Plain Language Summary**

An unintended pregnancy is a pregnancy that occurs either when the woman did not intend to get pregnant at all or when she intended to get pregnant but became pregnant at an inappropriate time. This type of pregnancy can have negative effects on the physical and mental health of women during pregnancy and after delivery. So far, no study has investigated the relationship between the experience of pregnancy and unintended pregnancy, and the existing studies, which are qualitative and based on interviews, have examined women's experiences of unintended pregnancy. For the first time, our study examined the relationship between pregnancy experience and unintended pregnancy in 488 women using a valid questionnaire, the Pregnancy Experience Scale, which includes two parts: uplifts and hassles specific to pregnancy. Our findings showed that women who have an unintended pregnancy have a worse pregnancy experience in both uplifts and hassles specific to pregnancy. Also, limited studies have investigated the prevalence of unintended pregnancy in Iran, especially in the city of Tabriz in recent years. Our study, found this prevalence to be 30.7% in Tabriz, with 24.3% of women having no intention of getting pregnant and 6.4% of women experiencing pregnancy at an inappropriate time. The recommendation is to implement strategies to reduce unintended pregnancy rates and improve women's knowledge of fertility, sexuality, and contraception.

# Background

An unintended pregnancy refers to a situation where a woman either did not intend to conceive at all or intended to conceive but at an inappropriate time [1-3]. Some risk factors associated with unintended pregnancy include the woman's age, wealth index, place of residence, number of children, intention to use contraception, living arrangements (alone or with a partner), age at the start of cohabitation, low socioeconomic status, history of previous abortions, and history of depression [4-7]. Globally, around 44% of pregnancies from 2010 to 2014 were unintended. The rate of unintended pregnancies is significantly higher in developing regions compared to developed ones [1]. In 2012, out of around 213 million pregnancies globally, 89% occurred in developing nations, with over half of these pregnancies being in Asia [2]. It has also been shown that from 2015 to 2019, there were 121 million unintended pregnancies annually, resulting in 64 unintended pregnancies per 1,000 women aged 15-49 [3]. The prevalence of unintended pregnancies in Iran has been reported differently in various studies, ranging from 17% in the city of Shiraz to 27.8% in the city of Arak [8, 9].

Unintended pregnancy, along with the consequent unsafe abortions, can impose significant financial and social burdens. These burdens can affect fertility rates, diminish the quality of life, and strain the public healthcare system [10-12]. Unintended pregnancy poses a challenging situation for women and public health in low- and middle-income countries, and it is associated with adverse maternal and infant outcomes [13]. These women are more at risk of cesarean section and excessive weight gain during pregnancy [14]. Women with unintended pregnancies are less likely to receive adequate prenatal care and are more likely to delay starting prenatal care. They also tend to use folic acid inappropriately and below the recommended amount. Additionally, they may continue to smoke during pregnancy and are less likely to breastfeed for 8 weeks or more [15, 16]. Unintended pregnancy is associated with low birth weight, preterm birth, as well as incomplete infant immunization [17–19]. These women may also experience complications during pregnancy such as preeclampsia, bleeding, and postpartum eclampsia [20].

Furthermore, women may resort to unsafe and illegal interventions, such as unsafe abortions, to terminate unintended pregnancies. These interventions can have detrimental effects on both the fetus and the mother's health, including complications such as peritonitis, uterine rupture, small bowel perforation, and fetal complications [21, 22]. Globally, it has been demonstrated that out of every 1000 women with unintended pregnancies, approximately 39 cases result in abortion [3].

Unintended pregnancy has varying impacts on women, and some individuals may experience more adverse consequences than others. It can lead to a decrease in the quality of relationships with partners and the ability to adapt to the work environment. Personal well-being and a negative outlook on life are mechanisms through which unintended pregnancies can have undesirable effects on a woman's life [23]. In addition, the negative experience of unintended pregnancy can lead to undesirable psychological disorders, including anxiety, post-traumatic stress disorder (PTSD), and postpartum depression [24, 25].

Some qualitative studies have been conducted in Iran [26–29], particularly among the Azari ethnicity [22], to examine women's experiences of unintended pregnancy. These studies consistently demonstrate the negative effects of such pregnancies on the physical and psychological health of mothers, including resorting to abortion as a coping mechanism and experiencing feelings of guilt afterward, fear of judgment from their family, friends,

community, or society reactions, and anxiety about taking on the role of motherhood. However, there has been a lack of quantitative studies investigating women's experiences of these pregnancies.

Given the negative impact of unintended pregnancy on health and quality of life, as well as the scarcity of quantitative studies exploring the relationship between pregnancy experience and this particular type of pregnancy, and the limited research on the prevalence of unintended pregnancy in Iran, including the absence of a study conducted in the city of Tabriz that considers both unwanted and mistimed pregnancies, the present study was undertaken with the following objectives:

- 1- To determine the prevalence of unintended pregnancy.
- 2- To identify the predictors of unintended pregnancy.
- To examine whether unintended pregnancy is a predictor of pregnancy uplifts.
- 4- To investigate whether unintended pregnancy is a predictor of pregnancy hassles.

# Methods

# Study setting and participants

The present study was a cross-sectional study conducted to determine the prevalence of unintended pregnancy and its impact on the pregnancy experience among 488 eligible pregnant women in healthcare centers in Tabriz City from February 2022 to January 2023.

The inclusion criteria for the study consisted of pregnant women between 20 to 40 weeks' gestation, having physical and mental health based on the mother's selfreport, the absence of any known diseases or abnormalities in the fetus based on the pregnant woman's medical records, being primiparous or multiparous and having a singleton pregnancy.

The exclusion criteria included women who were unwilling to participate, those who had experienced a significant adverse event in the past three months (such as the death of a family member or divorce), those who had incomplete responses to more than 20% of the questionnaire items, and those with a history of psychiatric illnesses (such as severe depression or requiring medication) based on the mother's self-report and medical records.

# Sample size

The sample size was determined based on the study conducted by Almasi-Hashiani, using the single proportion formula. Considering the prevalence of unintended pregnancy as 19.81%, an alpha ( $\alpha$ ) value of 0.05, and a study precision of 0.05, the initial sample size was calculated to be 244. However, since the sampling method used was cluster sampling with a design effect of 2, the final sample size was adjusted to 488 individuals [30].

$$N = \frac{(Z_{1-a/2})^2 * p(1-p)}{d^2}$$

# Data collection

After obtaining ethical approval from the Tabriz University of Medical Sciences (IR.TBZMED.REC.1400.681), permission for sampling was granted. Cluster sampling was employed in the study. Initially, one-fourth of the healthcare centers in Tabriz city (21 comprehensive health centers) were randomly selected using the website www.random.org. The researcher then visited the comprehensive health center and extracted a list of pregnant women between 20 and 40 weeks of gestation from the available records (SIB system). From the prepared list, the required number of samples for each center was determined proportionally and randomly selected. Subsequently, contact was made with these individuals via telephone, during which the research objectives and methods were briefly explained. During the phone call, individuals were assessed for inclusion and exclusion criteria, and if eligible and willing to participate in the research, they were invited to attend the designated center at the scheduled time. In the face-to-face meeting, the research objectives and methods were fully explained, and if pregnant women were willing to cooperate, the socio-demographic questionnaire and pregnancy experience scale (PES) were completed through interviews with the participants.

#### Data collection tools

This study collected data using the socio-demographic questionnaire, and the Pregnancy Experience Scale (PES).

# Socio-demographic questionnaire

This questionnaire includes questions about age, level of education, employment status of the woman and her spouse, income sufficiency, and unintended pregnancy (unwanted or mistimed). To determine the validity of this questionnaire, content, and face validity were assessed by the faculty members of the Tabriz University of Medical Sciences.

# Pregnancy Experience Scale (PES)

The PES, developed by Janet A. DiPietro in 2008, was utilized to assess the pregnancy experience. The questionnaire consists of 20 items, with 10 items dedicated to measuring uplifts specific to pregnancy and 10 items focused on hassles specific to pregnancy. Participants provided their responses on a four-point Likert scale, ranging from "Not at all" (score of zero) to "Very much" (score of three) for each item [31]. The validity and reliability of the Persian version of the pregnancy experience questionnaire were assessed in a study conducted by Ebadi et al. [32]. The overall tool demonstrated a reliability coefficient 0.714, indicating satisfactory internal consistency. Moreover, reliability coefficients of 0.777 and 0.672 were reported for the uplifts and hassles subscales, respectively, providing additional evidence of the questionnaire's reliability in measuring specific aspects of pregnancy experiences. The Cronbach's alpha coefficients calculated for the current study were 0.70 for the overall instrument, 0.82 for uplifts, and 0.73 for hassles specific to pregnancy. The Cronbach's alpha values of the original scale obtained for the uplifts and hassles subscales were 0.82 and 0.83, respectively.

# Data analysis

After collecting information from all the research participants, the data were analyzed using SPSS-Version 24 software. Descriptive statistics, including frequency (percentage) and mean (SD=standard deviation), were used to describe the socio-demographic characteristics and prevalence of unintended pregnancy. Binary logistic regression was used to identify the predictors of pregnancy desirability. Variables with p < 0.05 in the sociodemographic characteristics, as presented in Table 1, were included as independent variables, while pregnancy desirability was designated as the dependent variable in the binary logistic regression analysis. To examine the relationship between unintended pregnancy and pregnancy experience, an independent t-test was employed for bivariate analysis, and a general linear model (GLM) was utilized for multivariate analysis, with control of potential confounding variables.

#### Results

The prevalence of unintended pregnancies in Tabriz-Iran was 30.7%. Out of this percentage, 24.3% were identified as unwanted pregnancies, while 6.4% were mistimed pregnancies.

There was no significant difference in socio-demographic characteristics among participants with unintended and intended pregnancies except of age of the woman (p=0.001) and age of the husband (p=0.003) (Table 1).

The results of the binary logistic regression analysis indicated that the lower age of both the woman and her spouse were significant predictors for unintended pregnancy. Women aged under 19 years (adjusted odds ratio [aOR]: 2.91; 95% CI: 1.64–5.16; p < 0.001) and those aged 20–29 years (aOR: 1.97; 95% CI: 1.12–3.48; p = 0.019), as well as their spouses aged 19–29 years (aOR: 2.69; 95%

CI: 1.50–4.85; p=0.001), had a higher chance of experiencing unintended pregnancies compared to higher ages (Table 2).

Among women with unintended pregnancy, the mean ± SD of uplifts was  $14.56 \pm 5.53$ , and the mean ± SD of hassles was  $16.78 \pm 4.69$  (obtainable score range of 0 to 30). In contrast, among women with intended pregnancy, the mean ± SD of uplifts was  $19.56 \pm 4.80$ , and the mean ± SD of hassles was  $13.86 \pm 4.33$ , within the same score range of 0 to 30. Based on an independent t-test, the mean score of uplifts in women with unintended pregnancy was significantly lower than in women with intended pregnancy (mean difference (MD): -4.99; 95% confidence interval (CI): -5.96 to -4.02; p < 0.001), While the mean score of hassles in women with unintended pregnancy was significantly higher than women with intended pregnancy (MD: 2.92; 95% CI: 2.03 to 3.80; p < 0.001) (Table 3).

Based on bivariate analysis, the variables of women's age (p=0.018) and husband's education level (p=0.034) showed a relationship with uplifts. Additionally, women's educational level (p=0.026) and women's jobs (p=0.175) were found to be related to hassles, with a significance level of P < 0.2. These variables, along with the unintended pregnancy variable, were included as independent variables in the GLM, with hassles and uplifts serving as the dependent variables.

The findings indicated that women who experienced unintended pregnancies exhibited significantly lower scores for uplifts (B=-4.99; 95% CI: -5.96 to -4.03; P < 0.001) and higher scores for hassles (B=2.92; 95% CI: 2.06 to 3.78; P < 0.001) compared to women with intended pregnancies. Additionally, women with education below diploma level, specifically primary school (B=1.98; 95% CI: 0.58 to 3.39; P=0.006), secondary school (B=1.46; 95% CI: 0.02 to 2.91; P=0.046), and high school (B=2.38; 95% CI: 0.87 to 3.89; P=0.002) also obtained higher scores for hassles compared to those with university education (Table 4).

# Discussion

The results of the present study showed that approximately one-third of women in the city of Tabriz had unintended pregnancies. Furthermore, unintended pregnancy was identified as a predictive factor for a worse pregnancy experience.

In the current study, the prevalence of unintended pregnancy was found to be 30.7%, which is higher than the reported prevalence in countries such as Saudi Arabia (26.4%) and Ethiopia (28%) [33, 34]. Moreover, it is higher than the prevalence of unintended pregnancy reported in the latest study conducted in Iran, where in a cross-sectional study of 5152 pregnancies in 103 hospitals in

Variables	Relationship with pregnancy desirability			Relationship with pregnancy experience			
	Intended pregnancy ( <i>n</i> = 338) N (%)	Unintended pregnancy $(n = 150)$		Uplifts		Hassles	
		N (%)	P value	Mean (SD)	P value	Mean (SD)	P value
Women's age (Years)							
≥19	32 (21.3)	36 (10.7)	<b>0.001</b> <sup>a</sup>	16.92±5.33	0.018 <sup>⊂</sup>	$14.32 \pm 4.94$	0.076 <sup>c</sup>
20–29	50 (33.3)	164 (48.5)		18.91±5.16		$15.28 \pm 4.59$	
30–39	58 (38.7)	129 (38.2)		$17.42 \pm 5.80$		14.14±4.58	
≥40	10 (6.7)	9 (2.7)		$18.0 \pm 5.54$		$15.66 \pm 4.06$	
Husband's age (Years)							
19–29	29 (19.3)	110 (32.5)	<b>0.003</b> <sup>a</sup>	18.48±5.25	0.323 <sup>c</sup>	$14.02 \pm 4.56$	0.489 <sup>c</sup>
30–39	84 (56)	176 (52.1)		$18.0 \pm 5.58$		$15.13 \pm 4.54$	
≥40	37 (24.7)	52 (15.4)		17.35±5.83		$14.84 \pm 4.95$	
Women's educational leve	el						
Primary school	109 (32.2)	38 (25.3)	0.260 <sup>a</sup>	$18.38 \pm 5.60$	0.503 <sup>c</sup>	$15.14 \pm 4.54$	0.026 <sup>⊂</sup>
Secondary school	76 (22.5)	46 (30.7)		$17.40 \pm 5.28$		$14.62 \pm 4.65$	
High school	66 (19.5)	29 (19.3)		18.22±5.58		$15.53 \pm 4.66$	
Diploma	44 (13.0)	22 (14.7)		$17.56 \pm 5.80$		$14.46 \pm 4.45$	
University	43 (12.7)	15 (10.0)		$18.60 \pm 5.54$		$13.15 \pm 4.75$	
Husband's education leve	2						
Primary school	37 (10.9)	13 (8.70)	0.476 <sup>a</sup>	$16.36 \pm 5.06$	0.034 <sup>⊂</sup>	$14.22 \pm 4.65$	0.541 <sup>c</sup>
Secondary school	55 (16.3)	21 (14.0)		17.51±5.98		$14.53 \pm 4.42$	
High school	48 (14.2)	19 (12.7)		17.16±5.24		$15.34 \pm 5.22$	
Diploma	100 (29.6)	57 (38.0)		18.67±5.30		$15.05 \pm 4.34$	
University	98 (29.0)	40 (26.7)		18.58±5.72		$14.46 \pm 4.79$	
Women's Job							
Housewife	292 (86.4)	132 (88.0)	0.666 <sup>b</sup>	$17.95 \pm 5.54$	0.439 <sup>d</sup>	$14.88 \pm 4.53$	0.175 <sup>d</sup>
Employed	46 (13.6)	18 (12.0)		$18.53 \pm 5.545$		$13.93 \pm 5.26$	
Husband's job							
Worker	97 (28.7)	39 (26.0)	0.306 <sup>b</sup>	17.33±5.36	0.268 <sup>c</sup>	$14.64 \pm 4.51$	0.289 <sup>c</sup>
Employee	63 (18.6)	20 (13.3)		18.81±6.12		$14.15 \pm 4.80$	
Shopkeeper	18 (5.30)	7 (4.70)		18.08 (4.41)		13.92 (4.46)	
Self-employment	160 (47.3)	84 (56.0)		18.13 (5.52)		15.11 (4.66)	
Sufficiency of family inco	me						
Insufficient	18 (5.30)	13 (8.70)	0.390 <sup>a</sup>	18.44 (5.57)	0.467 <sup>c</sup>	14.83 (4.66)	0.826 <sup>c</sup>
Relatively sufficient	263 (77.8)	113 (75.3)		18.02 (5.49)		14.70 (4.66)	
Completely sufficient	57 (16.9)	24 (16.0)		17.0 (6.02)		15.22 (4.54)	

# Table 1 Socio-demographic characteristics and their relationship with pregnancy desirability and pregnancy experience

SD Standard Deviation, N Number

<sup>a</sup> Chi-square tests

<sup>b</sup> Fisher's exact tests

<sup>c</sup> One-way ANOVA

<sup>d</sup> Independent t-test

Tehran (the capital city) in 2015, it was considered to be 19.81% [30]. Furthermore, in a cross-sectional study conducted in 2014 in Kermanshah, West Iran, involving 248 women aged 15–49, the prevalence of unintended pregnancy was estimated to be 2.21% [35]. Additionally, in a survey conducted in 2013 in Shiraz City, the prevalence of unintended pregnancy was estimated to be 17% [4].

In another analytical cross-sectional study of 352 mothers with infants aged 6–12 months in 2007 in Arak City, the prevalence of unintended pregnancy was 8.27% [5]. This difference could be attributed to population growth policies in Iran in recent years and the fact that this study was conducted during the COVID-19 pandemic. In a cross-sectional study conducted in Iran to compare

**Table 2** Factors associated with pregnancy desirability based on binary logistic regression

Variable	OR	(95%Cl <sup>a</sup> )	P value
Women's age			
≥19	2.91	1.64-5.16	< 0.001
20–29	1.97	1.12-3.48	0.019
30–39	0.80	2.89-2.21	0.668
≤40 (Reference)	1		
Husband's age			
19–29	2.69	1.50-4.85	0.001
30–39	1.49	0.90-2.44	0.114
≤40 (Reference)	1		

<sup>a</sup> Confidence interval

contraceptive methods, abortion, and unintended pregnancies before and during the COVID-19 pandemic, it was found that the prevalence of unintended pregnancies increased from 20.4% before the pandemic to 25.4% during the pandemic [36].

According to a study [37], it has been found that women's reports of pregnancy planning during the postpartum period can lead to unreliable information. In a cross-sectional study conducted in Tabriz in 2020 by Taheri et al. [38], they reported a prevalence of 26.2% for unwanted pregnancy, while our current study reports a slightly lower prevalence of 24.3%. These differences in findings may be attributed to the evaluation of pregnancy planning during both the pregnancy and postpartum periods, as well as the inclusion of participants from diverse ethnic backgrounds in Taheri et al's study. In our present study, we specifically examined unwanted pregnancy during pregnancy among Azeri women.

Significant differences were observed in terms of uplifts and hassles between the two groups of pregnant women classified based on their intended and unintended pregnancies. Specifically, the group with unintended pregnancies demonstrated lower levels of uplifts and higher levels of hassles. However, it is important to note that there is currently a lack of quantitative research investigating this aspect. Several qualitative studies have demonstrated the negative effects of unintended pregnancy on women's pregnancy experience [22, 26, 27, 28, 29, 39, 40]. In a qualitative study conducted in Tabriz, Iran, Mohammadi et al. demonstrated that unintended pregnancy poses challenges for women and can result in social deprivation, physical and psychological risks, as well as unsafe and illegal consequences of abortion for both the mother and the infant [22]. Furthermore, several other qualitative studies conducted in Iran have also revealed that unintended pregnancy creates significant psychological pressures for women. Factors such as unpreparedness, both physically and emotionally, financial constraints in accepting the pregnancy, adherence to the two-child policy, feelings of embarrassment regarding pregnancy in the presence of other children and relatives, conflicting religious beliefs, and contemplation of fetal abortion are among the perceptions associated with unintended pregnancy among women [26-29]. Additionally, it should be noted that unintended pregnancy and low educational attainment in women can lead to psychological consequences, which, in turn, can impact the overall pregnancy experience [41].

Results showed that unintended pregnancy can serve as a predictor of both uplifts and hassles. The emotional impact of unintended pregnancy on women is significant, as it often leads to a sense of ambivalence and uncertainty. Furthermore, women experiencing unintended pregnancies tend to report lower levels of happiness [42]. The absence of prior pregnancy planning and the sudden shift in circumstances can contribute to heightened levels of anxiety among women [43]. However, it is important to note that personal resilience and individual circumstances can also influence the emotional outcomes of unplanned pregnancies [44].

The findings of the present study also highlight a significant positive correlation between women's educational level and the experience of hassles. Specifically, our results indicate that women with below high school education tend to report higher hassle scores. This observation aligns with previous research indicating that women with lower levels of education are more susceptible to adverse pregnancy outcomes, such as stillbirth [45].

Table 3 Comparison of pregnancy experience in participants with unintended or intended pregnancy

Pregnancy experience	Intended pregnancy N=338	Unintended pregnancy N=150	Unintended Mean difference (95%Cl) pregnancy $N = 150$		Obtainable score range	<i>P</i> value <sup>a</sup>
	Mean±SD	$Mean \pm SD$				
Uplifts	19.56±4.80	14.56±5.53	-4.99 (-5.96 to -4.02)	4–30	0–30	< 0.001
Hassles	13.86±4.33	16.78±4.69	2.92 (2.03 to 3.80)	3–28	0–30	< 0.001

CI confidence interval

<sup>a</sup> Independent t-test

**Table 4** Relationship of unintended pregnancy with pregnancy

 experience based on adjusted general linear model

Variable	B (95% Cl <sup>a</sup> )	P-value
Pregnancy experience (Uplifts)		
Unintended pregnancy		
Yes	-4.99 (-5.96 to -4.03)	< 0.001
No	0	
Husband's education level		
Primary school	0.37 (-1.42 to 2.17)	0.685
Secondary school	0.30 (-1.24 to 1.86)	0.697
High school	0.81 (-0.80 to 2.44)	0.323
Diploma	-0.31 (-1.58 to 0.96)	0.633
University	0	
Women's age		
≥19	-0.79 (-3.60 to 2.00)	0.577
20–29	1.16 (-1.42 to 3.75)	0.377
30–39	-0.28 (-2.89 to 2.31)	0.828
≥40	0	
Pregnancy experience (Hassles)		
Unintended pregnancy		
Yes	2.92 (2.06 to 3.78)	< 0.001
No	0	
Women's educational level		
Primary school	1.98 (0.58 to 3.39)	0.006
Secondary school	1.46 (0.023 to 2.91)	0.046
High school	2.38 (0.87 to 3.89)	0.002
Diploma	1.31 (-0.31 to 2.94)	0.114
University	0	
Women's Job		
Housewife	0.94 (-0.27 to 2.17)	0.127
Employed	0	

<sup>a</sup> Confidence interval

Moreover, educational attainment has been found to be associated with fertility and sexual knowledge, which in turn can influence the overall pregnancy experience [46, 47]. It is worth noting that attitudes towards fertility, education, and awareness play a crucial role in predicting the inclination towards future pregnancies, particularly in the context of the first and second child [48].

In the present study, we also found that a lower age of both the woman and her spouse was identified as a risk factor for unintended pregnancy. These findings are consistent with previous studies [4, 5, 49], which have similarly shown that a lower age is a predictor of unintended pregnancy. This observation may be attributed to limited access to high-quality sexual and reproductive health information among individuals in these age groups [49]. Limited access to education and family planning services presents a particular challenge for young and healthy women due to Iran's population growth policies [4, 23]. Additionally, the flexibility of the work environment for employed women is another influential factor in their experience of unintended pregnancy [23]. In response to these challenges, Iran has implemented new population policies that include strategic measures to support young and healthy women, such as providing postpartum leave with wages and the option to work from home. These provisions aim to facilitate a better balance between family and work life [50].

Furthermore, contrary to previous studies [51, 52], although the educational level of the woman was not identified as a risk factor for unintended pregnancy in the present study, lower education level of women was found to be a predictor of a worse pregnancy experience among women with unintended pregnancies.

It is necessary to consider interventional studies in light of Iran's population increase policy. These interventions include promoting the continuation of pregnancies, preventing abortions, educating women about the mental and physical risks associated with abortion, and facilitating a positive childbirth experience. It is also important to provide counseling services to women and their families, as well as educate and encourage them about the importance of parenthood and having children to strengthen families. Furthermore, it would be essential to quantitatively evaluate the impact of such interventions on the pregnancy experience. Additionally, we recommend conducting mixed-method studies to obtain comprehensive data on unintended pregnancy experiences while considering the proposed interventions.

#### Strengths and limitations

The strengths of the current study include the quantitative investigation of the relationship between pregnancy experience and unintended pregnancy for the first time. This study also utilized standardized tools to measure the pregnancy experience. Additionally, the study employed a random selection method to choose participants from healthcare centers in Tabriz city.

The current study has several limitations. Firstly, as a cross-sectional study, it cannot establish a causal relationship between pregnancy experience and unintended pregnancy. Future research using longitudinal designs is needed to address this limitation. Additionally, the generalizability of the findings may be limited due to the focus on healthy singleton pregnancies and a specific ethnic group (Azari ethnicity), which restricts the representation of diverse populations. Furthermore, the study did not include women with pregnancies below 20 weeks' gestational age, multiple pregnancies, women with mental disorders and related medication use, or those who recently experienced significant

adverse events. Future research should aim to incorporate these subgroups to ensure a more comprehensive understanding of the experiences of unintended pregnancy.

# Conclusions

The study findings indicated a significantly high prevalence of unintended pregnancies among women in the city of Tabriz. The high prevalence of unintended pregnancies, along with the observed differences in the experiences of uplifts and hassles, highlights the pressing need for targeted interventions. Future studies should take into account the broader policy framework, such as Iran's population increase policy, and address the unique challenges faced by women with unintended pregnancies. This approach will enable the development of contextspecific interventions that effectively meet the needs of this specific population.

#### Abbreviations

- AOR Adjusted odds ratio
- CI Confidence interval
- GLM General linear model
- MD Mean difference
- PES Pregnancy experience scale
- SD Standard deviation

#### Acknowledgements

We appreciate the participation of women in this study, as well as the collaboration of comprehensive health center in Tabriz city in conducting this research. We also acknowledge the financial support provided by Tabriz University of Medical Sciences.

#### Authors' contributions

The authors made the following contributions to this study: MMi, MMa, RN, and MN were responsible for the study's conception and design. MMa and RN conducted the data collection. MMa, RN, and MN prepared the initial draft of the paper and the final manuscript. MMi performed the data analysis. All authors thoroughly reviewed and approved the final manuscript for publication.

#### Funding

The study received funding from Tabriz University of Medical Sciences, Tabriz-Iran.

#### Availability of data and materials

Upon request, the corresponding author will provide access to the data.

## Declarations

# Ethics approval and consent of participants

All Helsinki Declaration principles were followed in conducting the present study. After obtaining ethical approval from the committee (IR.TBZMED. REC.1400.681), sampling permission was obtained from Tabriz University of Medical Sciences. Informed consent was obtained in writing from all participants, and necessary measures were taken to ensure the confidentiality of participants' information. Participants were assured of confidentiality, and their privacy was respected throughout the study.

#### **Consent for publication**

Not applicable.

#### **Competing interests**

The authors declare no competing interests.

#### Author details

<sup>1</sup>Student Research Committee, Midwifery Department, Faculty of Nursing and Midwifery, Tabriz University of Medical Sciences, Tabriz, Iran. <sup>2</sup>Tabriz University of Medical Sciences, Tabriz, Iran. <sup>3</sup>Medical School, Ardabil University of Medical Sciences, Ardabil, Iran. <sup>4</sup>Social Determinants of Health Research Center, Tabriz University of Medical Sciences, Tabriz, Iran.

#### Received: 4 January 2024 Accepted: 29 May 2024 Published online: 05 June 2024

#### References

- Bearak J, Popinchalk A, Alkema L, Sedgh G. Global, regional, and subregional trends in unintended pregnancy and its outcomes from 1990 to 2014: estimates from a Bayesian hierarchical model. Lancet Glob Health. 2018;6(4):e380–9.
- Sedgh G, Singh S, Hussain R. Intended and unintended pregnancies worldwide in 2012 and recent trends. Stud Fam Plann. 2014;45(3):301–14.
- Bearak J, Popinchalk A, Ganatra B, Moller AB, Tunçalp Ö, Beavin C, Kwok L, Alkema L. Unintended pregnancy and abortion by income, region, and the legal status of abortion: estimates from a comprehensive model for 1990–2019. Lancet Glob Health. 2020;8(9):e1152–61.
- Iseyemi A, Zhao Q, McNicholas C, Peipert JF. Socioeconomic status as a risk factor for unintended pregnancy in the contraceptive CHOICE Project. Obstet Gynecol. 2017;130(3):609–15.
- Sarder A, Islam SMS, Maniruzzaman, Talukder A, Ahammed B. Prevalence of unintended pregnancy and its associated factors: evidence from six south Asian countries. Plos One. 2021;16(2):e0245923.
- Fite RO, Mohammedamin A, Abebe TW. Unintended pregnancy and associated factors among pregnant women in Arsi Negele Woreda, West Arsi Zone, Ethiopia. BMC Res Notes. 2018;11(1):671.
- Hall KS, Richards JL, Harris KM. Social disparities in the relationship between depression and unintended pregnancy during adolescence and young adulthood. J Adolesc Health. 2017;60(6):688–97.
- Asadi Sarvestani K, Ahmadi A, Enayat H, Movahed M. Level and factors related to unintended pregnancy with a brief review of new population policies in Iran. Iran J Public Health. 2017;46(7):973–81.
- 9. Beygi AM, Salehi NM, Bayati A. Prevalence of unintended pregnancy and its related factors in Arak 2007. Koomesh. 2009;10(3):Pe201–6.
- Yazdkhasti M, Pourreza A, Pirak A, Abdi F. Unintended pregnancy and its adverse social and economic consequences on health system: a narrative review article. Iran J Public Health. 2015;44(1):12–21.
- Sonfield A, Kost K, Gold RB, Finer LB. The public costs of births resulting from unintended pregnancies: national and state-level estimates. Perspect Sex Reprod Health. 2011;43(2):94–102.
- Naji Abhary M, Shaghaghi F, Safinejad H, Asadi L, Dehnavi Mohebbi Z. The effects of unsafe abortion on the health system from a socioeconomic perspective: a narrative review. J Midwifery Reprod Health. 2023;11(4):3896–909.
- Khan MN, Islam MM. Women's experience of unintended pregnancy and changes in contraceptive methods: evidence from a nationally representative survey. Reprod Health. 2022;19(1):187.
- Omani-Samani R, Ranjbaran M, Mohammadi M, Esmailzadeh A, Sepidarkish M, Maroufizadeh S, Almasi-Hashiani A. Impact of unintended pregnancy on maternal and neonatal outcomes. J Obstet Gynaecol India. 2019;69(2):136–41.
- Abame DE, Abera M, Tesfay A, Yohannes Y, Ermias D, Markos T, Goba G. Relationship between unintended pregnancy and antenatal care use during pregnancy in Hadiya Zone Southern Ethiopia. J Reprod Infertil. 2019;20(1):42–51.
- Cheng D, Schwarz EB, Douglas E, Horon I. Unintended pregnancy and associated maternal preconception, prenatal and postpartum behaviors. Contraception. 2009;79(3):194–8.
- Rahman M, Nasrin SO, Rahman M, Rahman A, Mostofa G, Jesmin SS, Buchanan F. Maternal pregnancy intention and its association with low birthweight and pregnancy complications in Bangladesh: findings from a hospital-based study. Int Health. 2019;11(6):447.

- Singh A, Singh A, Thapa S. Adverse consequences of unintended pregnancy for maternal and child health in Nepal. Asia Pac J Public Health. 2015;27(2):NP1481-91.
- Wagan F, Siyal AA, Ali R, Taqi T. Major consequences, determinants and obstetrical outcomes of unintended pregnancy. Natl Editorial Advisory Board. 2018;29(11):43–7.
- Dehingia N, Dixit A, Atmavilas Y, Chandurkar D, Singh K, Silverman J, Raj A. Unintended pregnancy and maternal health complications: cross-sectional analysis of data from rural Uttar Pradesh, India. BMC Pregn Childb. 2020;20(1):188.
- Rashidpouraie R, Dastjerdi MV, Shojaei A, Saeeditehrani S, Sharifi M, Joodaki K, Moosavinejad S, Rashidpouraei M, Zahedi L, Mashkoori A, Larijani B. Complications of illegal abortion in the suburbs of Tehran: a 9-year cross-sectional study. J Res Med Sci. 2021;26:89.
- Mohammadi E, Nourizadeh R, Simbar M. Iranian Azeri women's perceptions of unintended pregnancy: a qualitative study. Iran J Nurs Midw Res. 2015;20(2):255–62.
- Kavanaugh ML, Kost K, Frohwirth L, Maddow-Zimet I, Gor V. Parents' experience of unintended childbearing: a qualitative study of factors that mitigate or exacerbate effects. Soc Sci Med. 2017;174:133–41.
- 24. Judge-Golden CP, Borrero S, Zhao X, Mor MK, Callegari LS. The association between mental health disorders and history of unintended pregnancy among women veterans. J Gen Intern Med. 2018;33(12):2092–9.
- Qiu X, Zhang S, Sun X, Li H, Wang D. Unintended pregnancy and postpartum depression: a meta-analysis of cohort and case-control studies. J Psychosom Res. 2020;138: 110259.
- Akbarzadeh M, Yazdanpanahi Z, Zarshenas L, Sharif F. The women's perceptions about unwanted pregnancy: a qualitative study in Iran. Glob J Health Sci. 2015;8(5):189–96.
- Mortazavi F, Damghanian M, Mottaghi Z, Shariati M. Women's experiences of unwanted pregnancy. J Kermanshah Univ Med Sci. 2012;15(6): e78904.
- Dehghan-Nayeri N, Tajvidi M. Experiences of pregnancy among Iranian adolescents: a qualitative study. Iran J Nurs Midw Res. 2014;19(7 Suppl 1):S7–12.
- Bayrami R, Taghipour A, Ebrahimipour H. Experience of unplanned pregnancy in women attending to health centers of Mashhad, Iran: a phenomenological study. Iran J Obstet Gynecol Infertil. 2014;16(87):15–23.
- Almasi-Hashiani A, Omani-Samani R, Sepidarkish M, Khodamoradi F, Ranjbaran M. Unintended pregnancy in Iran: prevalence and risk factors. Int J Women's Health Reprod Sci. 2019;7(3):319–23.
- DiPietro JA, Christensen AL, Costigan KA. The pregnancy experience scale-brief version. J Psychosom Obstet Gynaecol. 2008;29(4):262–7.
- Ebadi A, Kariman N, Hajifoghaha M. Psychometric properties and validation of the Persian translation of the pregnancy experience scale – brief version (PES). J Rafsanjan University Med Sci. 2017;15(9):809–20.
- Alsafar FA, Alnaeem LS, Almzraq LA, Alshouq ST, Albukhaytan WA. Prevalence of unplanned pregnancy and its psychological effect among pregnant females in Eastern Province, Saudi Arabia. IJMDC. 2022;6(10):1251– 60. https://doi.org/10.24911/IJMDC.51-1661468534.
- 34. Alene M, Yismaw L, Berelie Y, Kassie B, Yeshambel R, Assemie MA. Prevalence and determinants of unintended pregnancy in Ethiopia: A systematic review and meta-analysis of observational studies. PLoS One. 2020;15(4):e0231012.
- Shahbazin S, Gholamy A. Prevalence of unintended pregnancy and its related factors in Kermanshah, Kangavar city (west Iran). J Community Health Res. 2015;4(1):19–28.
- Rezaei F, Amiri-Farahani L, Haghani S, Pezaro S, Behmanesh F. The impact of the COVID-19 pandemic on contraceptive methods, abortion, and unintended pregnancy: a cross-sectional study. BMC Womens Health. 2023;23(1):357.
- Chamberlin S, Njerenga S, Smith-Greenaway E, Yeatman S. Women's life experiences and shifting reports of pregnancy planning. Matern Child Health J. 2022;26(8):1719–26.
- Taheri A, Vakili M, Mirzaei M, Modarresi, M: Frequency of unwanted pregnancies and its related factors in Tabriz hospitals in 2020. Iran J Obstet Gynecol Infertil. 2022;25(1):38–45.
- Mumah JN, Mulupi S, Wado YD, Ushie BA, Nai D, Kabiru CW, Izugbara CO. Adolescents' narratives of coping with unintended pregnancy in Nairobi's informal settlements. PLoS One. 2020;15(10):e0240797.

- Smith W, Turan JM, White K, Stringer KL, Helova A, Simpson T, Cockrill K. Social norms and stigma regarding unintended pregnancy and pregnancy decisions: A qualitative study of young women in Alabama. Perspect Sex Reprod Health. 2016;48(2):73–81.
- Ture P, Dambhare DG, Mundra A, Raut AV, Maliye CH, Deshmukh PR, Gupta SS, Bharambe MS, Garg BS. Magnitude and determinants of psychological morbidities among pregnant women: Results from a pregnancy cohort in rural Central India. Med J Armed Forces India. 2024;80(2):210–16.
- 42. Barton K, Redshaw M, Quigley MA, Carson C. Unplanned pregnancy and subsequent psychological distress in partnered women: a cross-sectional study of the role of relationship quality and wider social support. BMC Pregnancy Childbirth. 2017;17(1):44.
- Bahk J, Yun SC, Kim YM, Khang YH. Impact of unintended pregnancy on maternal mental health: a causal analysis using follow up data of the Panel Study on Korean Children (PSKC). BMC Pregnancy Childbirth. 2015;15:85.
- Olajubu A, Omoloye G, Olajubu T, et al. Stress and resilience among pregnant teenagers in Ile-Ife, Nigeria. Eur J Midwifery. 2021;5(3):1–9. https:// doi.org/10.18332/ejm/134181.
- 45. Li L, Wu Y, Yang Y, Wu Y, Zhuang Y, You D. Maternal educational inequalities about adverse pregnancy outcomes observed in a rural area of a province of China during a time period (2010-2018). J Epidemiol Community Health. 2022; jech-2021-217754.
- Altman MR, Oseguera T, McLemore MR, Kantrowitz-Gordon I, Franck LS, Lyndon A. Information and power: Women of color's experiences interacting with health care providers in pregnancy and birth. Soc Sci Med. 2019;238:112491. https://doi.org/10.1016/j.socscimed.2019.112491.
- Fegita P, Hikmah M, Malik R. Relationship between education level, age, and knowledge of pregnant women with antenatal care status. Sci J. 2022;1(2):154–64.
- Guzzo KB, Hayford SR. Adolescent reproductive attitudes and knowledge effects on early adult unintended and nonmarital fertility across gender. Adv Life Course Res. 2021;50:100430.
- Ikamari L, Izugbara C, Ochako R. Prevalence and determinants of unintended pregnancy among women in Nairobi, Kenya. BMC Pregnancy Childbirth. 2013;19;13:69.
- Mehri N, Messkoub M, Kunkel S. Trends, determinants and the implications of population aging in Iran. Ageing Int. 2020;45(4):327–43.
- Kebede KM, Belay AS, Shetano AA. Prevalence and determinants of unintended pregnancy in Ethiopia: narrative synthesis and meta-analysis. Heliyon. 2021;7(9):e07869.
- 52. Shankar M, Hooker L, Edvardsson K, Norman WV, Taft AJ. The prevalence and variations in unintended pregnancy by socio-demographic and health-related factors in a population-based cohort of young Australian women. Aust N Z J Public Health. 1. 2023;47(3):100046.

# **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.